Author's response to reviews

Title: Genetic association study of selected candidate genes (ApoB, LPL, Leptin) and telomere length in obese and hypertensive individuals.

Authors:

Birajalaxmi Das (birajalaxmi@yahoo.co.in)
Nilambari Pawar (nilambari.pawar@gmail.com)
Divyalakshmi Saini (ndivya@barc.gov.in)
M Seshadri (msesh@barc.gov.in)

Version: 11 Date: 21 September 2009

Author's response to reviews:

Dear Sir,

Few typographical errors have been corrected. In figure 2, th T/S ratio should be in parantese. "(T/S ratio" in Y axis was corrected as "T/S ratio).  

Now the manuscript proof copy is redy for publication.

Dr. Birajalaxmi Das

birajalaxmi@yahoo.co.in